Author's response to reviews

Title: Exome sequencing helped the fine diagnosis of two siblings afflicted with Timothy syndrome (TS2)

Authors:

Sebastian Fröhler (Sebastian.Froehler@mdc-berlin.de)
Moritz Kieslich (Moritz.Kieslich@charite.de)
Claudia Langnick (Claudia.Langnick@mdc-berlin.de)
Mirjam Feldkamp (Mirjam.Feldkamp@mdc-berlin.de)
Bernd Opgen-Rhein (Bernd.Opgen-Rhein@charite.de)
Felix Berger (Felix.Berger@charite.de)
Joachim C Will (Achim.Will@charite.de)
Wei Chen (Wei.Chen@mdc-berlin.de)

Version: 2 Date: 18 June 2013

Author's response to reviews:

Dear Editor,

We have changed the article type to 'Case Report' as suggested.

Written informed consent for the studies in our manuscript as well as for publishing this case was obtained from each family member before starting the study.

We could provide you with the consent forms if necessary over a secure channel (they are in German though).

Best regards

Sebastian Fröhler